Author’s response to reviews

Title: Syphilis and HIV prevalence and associated factors to their co-infection, hepatitis B and hepatitis C viruses prevalence among female sex workers in Rwanda

Authors:
Mwumvaneza Mutagoma (mutagoma@gmail.com)
Laetitia Nyirazinyoye (lnyirazi@gmail.com)
Dieudonné Sebuhoro (tontosebu@gmail.com)
David Riedel (davidjriedel@gmail.com)
Joseph Ntaganira (jntaganira@ur.ac.rw)

Version: 4 Date: 19 Jul 2017

Author’s response to reviews:

Dear Editorial team,

Comments from reviewers are responded

1. Please add a “Conclusions” section after the “Discussion” section.

R/ Conclusion section was added in the current version

2. Please amend the statement in the "Consent for publication" section, as this refers to consent obtained to publish potentially-identifying information (of patients/participants).

R/ The section was modified
3. Please include a legend for Table 3, as this seems to be missing.

R/ Legend were omitted. They are added in the final version